Author’s response to reviews

Title: The clinical practice guideline Palliative care for children and other strategies to enhance shared decision-making in pediatric palliative care; pediatricians’ critical reflections

Authors:

D. Dreesens (dunja.dreesens@maastrichtuniversity.nl; dunja.dreesens@gmail.com)

Lotte Veul (lotte.veul@gmail.com)

Jonne Westermann (j.westermann@ggzcentraal.nl)

Nicole Wijnands (nwijnands@outlook.com)

Leontien Kremer (l.c.kremer@amc.uva.nl)

Trudy van der Weijden (trudy.vanderweijden@maastrichtuniversity.nl)

Eduard Verhagen (a.a.e.verhagen@umcgroningen.nl)

Version: 1 Date: 11 Oct 2019

Author’s response to reviews:

Thank you for the useful comments.
We have listed our responses to the comments from the editor and reviewers in the revision note. When the comments have led to changes in the manuscript, we have included these parts in the revision note.

We have re-analyzed our findings and agree that in some places we might have been generalizing our results too much and we have now used more strict formulations.
Also, we have reformulated our conclusion. In doing so, we hope to meet the comments made.
We also looked at the other remarks made by you and the reviewers and addressed these in the manuscript. Furthermore, we have re-read the manuscript to improve clarity and grammar. These changes are shown in the version of the manuscript by means of track changes.